EO-1 Weekly Status Week of March 13 – March 19, 2008
Day of Year 073 - 079
Mission Day 2677 - 2683
Earth Observing-One (EO-1) General

There were 106 Data Collection Events (DCEs) scheduled this week.  
On March 18, a telecon was held with the International Federation Red Cross (IFRC) and other international organizations working in the area of worldwide flooding disaster alerts.  The following significant items were discussed:

· The NASA Global Model for flood potential forecast is updated every three hours for the entire globe.
· Hal Pierce (GSFC/Laboratory for Atmospheres) will provide answers to the following:
· Can the Global Model output be overlaid onto a Web Map Service (WMS)?

· Can the Global Model output be converted into a KML format for display in Google Earth?

· The International Federation Red Cross (IFRC) is also interested in potential flooding caused by snow melts in Central Asia.
INSTRUMENTS
All instruments operated nominally this week.
EO-1 Spacecraft Subsystems
Command and Data Handling (C&DH)

· Continued to experience problems playing back engineering data from its Solid-State-Recorders.
Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

A teleconference was held between the Advanced Information System Technology (AIST) Sensor Web collaborators on Thursday, March 13.  The following items of interest were discussed:
· Level 1G data from the February 21 Chincoteaque Bay image for a Rhodamine dye detection test was delivered to Troy Ames.  Level 1R software is being re-hosted as a Web Processing Service (WPS).  When completed Level 1R data can be produced by a WPS.  It will be delivered to Troy Ames and to the Defense Intelligence Agency (DIA) National Signatures Program for development of a Rhodamine dye detection algorithm.
· Still awaiting authorization from JPL to make their WPS openly available without needing an account and password.  This will make running WPS’s much easier.
· There has been some confusion on which metadata fields contained in the Level 1G GeoBliki metadata file to use during a KML conversion.  This confusion was cleared up in a subsequent telecom held on March 18.  However, a request was made to place an order at the USGS EROS facility for the same Level 1G product to compare the geolocation occurring between the two products.

· Members of the GSFC team are preparing to attend the OWS-5 demonstration that is part of the OGS technical conference being held in St. Louis the week of March 24-28.  The team has spent part of this preparation time coordinating the various views of EO-1 web services to make sure a consistent picture is presented.
· For OWS-5, all outstanding workflow compatibility issues have been resolved.

· OWS-5 video preparation is in process for delivery on March 21.

· A meeting with the SERVIR team, Panama Emergency Management Department, officials from Kenya, and EO-1 team is set for April 28.

Autonomous Sciencecraft Experiment (ASE)

· The ASE controlled EO-1 all week with the exception of the events noted below in Real-Time.
EO-1 MISSION OPERATIONS CENTER
Real-Time
No change from last report.
Mission Planning
· MOPSS is currently not functional on Speedwheel and Axle. 

· Database and System administrators are working closely to troubleshoot this problem.

· It appears that certain parts of the system did not handle the spring time change to daylight savings time properly.

· This timing discrepancy within the system is causing MOPSS software to fail at startup. 

Flight Dynamics
· Modified lunar calibration.

· FOT has changed the pitch offset values for the ALI and HSI instrument.

· Mission scientists have been working with the FOT to better target the moon.

· Instrument pointing offsets have been modified in the flight dynamics planning scripts to allow for improved targeting.

Data Analysis and Trending
· No change from last report.
System Administration
· No change from last report.
Operational Enhancements Requests and Discrepancy Reports
· New automation concept to deliver ASPEN products to the MOC.
· Proposed set of changes to the SGM code are to be implemented by the FOT.

· On March 19, the FOT was unsuccessful in trying to duplicate e-mail dropout scenarios with the original code to ensure that, with the modified code in place, it could be verified that the duplicate e-mail dropout scenarios problem was fixed. 
· As a result, the original SGM code has been retained and work will continue on identifying the source of the dropout scenarios problem.
GROUND AND SPACE NETWORK

Station Downtime
· No change from last report.
Operational Discrepancies
· No major problems to report.
Support Enhancements  
· No change from last report.
Upcoming Events
· Two lunar calibrations to be performed on March 22, 2008.

· Nominal lunar calibration to be performed during the 02:45z – 03:17z umbra.

· Modified lunar calibration to be performed during the 10:59z – 11:27z umbra.

· FOT has changed the pitch offsets in the flight dynamics planning scripts to improve targeting.

Imagery Status
Scenes and Engineering Cals planned for week of March 13 – March 19, 2008              106
Total scenes and engineering calibrations planned for entire mission – approximately 38,899
Total Scenes:  ALI scenes in the Level 0 archive            35,330 (as of March 19, 2008)

                        Hyperion scenes in the Level 0 archive    35,079
Publications and Presentations Status (as of 12-05-07)
404 publications 

284 external presentations

53 articles and press releases
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